Author's response to reviews

Title: A portrait of single and multiple HPV-type infections in Brazilian women of different age strata with squamous or glandular cervical lesions

Authors:

LEANDRO SA RESENDE (leandroresende@yahoo.com.br)
SILVIA H RABELO-SANTOS (rabelo.silvia@gmail.com)
LUIS O SARIAN (luis.sarian@gmail.com)
ROSANE RF ALVES (rosanefalves@gmail.com)
ANDREA ALVES RIBEIRO (andrea.ribeiro13@gmail.com)
LUIZ C ZEFERINO (zeferino@fcm.unicamp.br)
SOPHIE FM DERCHAIN (derchain@fcm.unicamp.br)

Version: 5 Date: 26 March 2014

Author's response to reviews:

We have now provided the sections that were missing in the original submission: authors contributions, acknowledgements, the names of the ethics committees, conflicts of interest statement.